Local production of 17β-oestradiol in the endometrium during the implantation window: a pilot study

L B P M Stevens Brentjens⁽¹⁾^{1,2}, D Obukhova^{2,3}, B Delvoux^{1,2}, J E den Hartog^{1,2}, B N Bui⁴, F Mol⁵, J P de Bruin⁶, D Besselink⁷, G Teklenburg⁸, F Morgan⁹, M Baker⁹, F J M Broekmans⁴, R J T van Golde^{1,2}, M Zamani Esteki^{2,3,10} and A Romano⁽¹⁾^{1,2}

¹Department of Obstetrics and Gynaecology, Maastricht University Medical Centre+, Maastricht, The Netherlands ²GROW School for Oncology and Reproduction, Maastricht University, Universiteitssingel, Maastricht, The Netherlands

³Department of Clinical Genetics, Maastricht University Medical Centre (MUMC+), Maastricht, The Netherlands

⁴Department of Gynaecology & Reproductive Medicine, University Medical Centre Utrecht, Heidelberglaan, Utrecht, The Netherlands

⁵Centre for Reproductive Medicine, Reproduction and Development, Amsterdam University Medical Centre, University of Amsterdam, Meibergdreef, Amsterdam, The Netherlands

⁶Department of Obstetrics and Gynaecology, Jeroen Bosch Hospital, Henri Dunantstraat, Hertogenbosch, The Netherlands

⁷Department of Obstetrics and Gynaecology, Radboud University Medical Centre, Geert Grooteplein Zuid, Nijmegen, The Netherlands ⁸Isala Fertility Clinic, Isala Hospital, Dokter van Heesweg, Zwolle, The Netherlands

⁹Department of Complex Tissue Regeneration, MERLN Institute, Maastricht University, Maastricht, The Netherlands

¹⁰Division of Obstetrics and Gynaecology, Department of Clinical Science, Intervention & Technology (CLINTEC), Karolinska Institutet, Stockholm, Sweden

Correspondence should be addressed to L B P M Stevens Brentjens: linda.brentjens@mumc.nl



https://raf.bioscientifica.com https://doi.org/10.1530/RAF-23-0065 © 2023 the author(s) Published by Bioscientifica Ltd



This work is licensed under a Creative Commons dealtrificeRonFiveAccomminational NoDenVaRUse4/2024 03:37:16PM cqffernationalOppense. Licensed under a Creative Commons nCommercial-NoDerivatives 4.0 International License. https://creativecommons.org/licenses/by-nc-nd/4.0/

Abstract

Sex steroids are converted to bioactive metabolites and vice versa by endometrial steroid-metabolising enzymes. Studies indicate that alterations in this metabolism might affect endometrial receptivity. This pilot study determined whether the endometrial formation and inactivation of 17β -oestradiol differed between the supposedly embryo-receptive endometrium and non-receptive endometrium of women undergoing IVF/intracytoplasmic sperm injection (ICSI). Endometrial biopsies were obtained from IVF/ICSI patients 5–8 days after ovulation in a natural cycle, prior to their second IVF/ICSI cycle with fresh embryo transfer (ET). Endometrial biopsies from patients who achieved clinical pregnancy after fresh ET (n = 15) were compared with endometrial biopsies from patients that did not conceive after fresh ET (n = 15). Formation of 17β-oestradiol (oxidative 17β-hydroxysteroid dehydrogenases (HSDs)), oestrone (reductive HSD17Bs) and inhibition of HSD17B1 activity were determined by high-performance liquid chromatography. The endometrial transcriptome was profiled using RNA sequencing followed by principal component analysis and differentially expressed gene analysis. The false discovery rate-adjusted P < 0.05 and log fold change >0.5 were selected as the screening threshold. Formation and inactivation of 17β-oestradiol resulted similar between groups. Inhibition of HSD17B1 activity was significantly higher in the non-pregnant group when only primary infertile women (n = 12) were considered (27.1%, n = 5vs 16.2%, n = 7, P = 0.04). Gene expression analysis confirmed the presence of HSD17B1 (encoding HSD17B1), HSD17B2 (encoding HSD17B2) and 33 of 46 analysed steroid metabolising enzymes in the endometrium. In the primary infertile subgroup (n = 10) 12 DEGs were found including *LINC02349* which has been linked to implantation. However, the exact relationship between steroid-metabolising enzyme activity, expression and implantation outcome requires further investigation in larger, well-defined patient groups.

Lay summary

Sex hormones are produced and broken down by enzymes that can be found in the endometrium (the inner lining of the womb). This enzyme activity might influence the chances of becoming pregnant. We compared (i) enzyme activity in the endometrium of 15 women who did and 15 women who did not become pregnant in their second *in vitro* fertilisation attempt, (ii) how enzyme activity can be blocked by an inhibitor, and (iii) differences in gene expression (the process by which instructions in our DNA are converted into a product). Enzyme activity was similar between groups. We found that in women who have never been pregnant in the past, inhibition of enzyme activity was higher and found differences in a gene that has been linked to the implantation of the embryo, but future studies should be performed in larger, well-defined patient groups to confirm these findings.

Keywords: ► oestrogen ► implantation ► HSD17B1 ► endometrium ► receptivity ► IVF

Reproduction and Fertility (2023) 4 e230065

Introduction

In Europe, pregnancy rates after IVF have reached a plateau during the last years, with average pregnancy chances around 33% per embryo transfer for both IVF and intracytoplasmic sperm injection (ICSI) (European IVF Monitoring Consortium *et al.* 2022). For implantation to occur, the endometrium must undergo a series of molecular and cellular events that would lead to a receptive endometrial state. The period in which the endometrium is receptive to an embryo, or window of implantation (WOI), is limited to days 19–24 of a regular 28-day menstrual cycle and is modulated by steroid hormones.

In particular, 17β -oestradiol is one of the fundamental regulators of endometrial functions and receptivity. During the follicular phase, it induces endometrial cell proliferation and primes the endometrium to respond to progesterone during the luteal phase. During the luteal phase, serum 17β -oestradiol levels decline, and intracellular 17β -oestradiol signalling is blunted to allow for full activation of progesterone signalling, which in turn induces stroma cell decidualisation (DeMayo & Lydon 2020).

Although 17β -oestradiol is mainly produced within the ovarian follicles, its production also occurs



in peripheral tissues such as fat, breast, liver, brain and endometrium, a process known as local steroid metabolism or intracrinology (Labrie 1991). The endometrium expresses several enzymes responsible for the generation of steroids and 17β-oestradiol, and is therefore capable of regulating local 176-oestradiol concentrations. As a consequence, 17^β-oestradiol tissue levels may differ from serum levels, which may lead to altered intracellular signalling activation (Konings et al. 2018a). The enzymes 17β-hydroxysteroid dehydrogenases (HSD17Bs) are particularly relevant in such local steroid metabolism. The redox balance between reductive and oxidative HSD17Bs controls the balance between the active 17^β-oestradiol and the less active oestrogen oestrone. Out of the 15 known HSD17Bs that have been characterised to date, the enzyme HSD17B1 is considered the most relevant for local 17β -oestradiol generation, as it has the highest affinity towards oestrogens, and catalyses the reduction of oestrone to 17β-oestradiol. The enzyme HSD17B2 catalyses the oxidation of 17β-oestradiol to oestrone. Besides HSD17B1 and 2, other reductive and oxidative HSD17Bs (i.e. HSD17B5, 6, 7, 8 and 14) might be engaged in the local oestrogen metabolism, but their role and impact in vivo are not fully elucidated (Konings et al. 2018a).

The role of 17β-oestradiol, oestrogen signalling and the influence of its metabolism during implantation is currently unclear. Previous studies on mice and human stroma cells in vitro indicate that local formation of 17β-oestradiol is fundamental for decidualisation of stroma cells (Das et al. 2009, Gibson et al. 2013). Altered regulation of local steroid metabolism by HSD17B enzymes has been linked to endometrial development throughout different stages of the menstrual cycle and to disorders such as endometriosis. Although other enzymes controlling local steroid metabolism have also been related with implantation in humans, there are no reports of HSD17B1 expression being associated with it. However, one study on ectopic pregnancies detected HSD17B1 expression in epithelial cells of the fallopian tube at the nidation site, suggesting a regulative role of this enzyme at the site of implantation (Li et al. 2003).

In the present study, we explored the redox balance of reductive and oxidative HSD17Bs in relation to endometrial receptivity during the window of implantation. We (i) compared the activity of reductive and oxidative HSD17B enzymes in mid-luteal endometrial biopsies in women who did or did not become pregnant in the IVF/ICSI cycle following the biopsy, (ii) assessed the engagement of HSD17B1 using a specific inhibitor and (iii) profiled the endometrial transcriptome to gain further insight into the molecular mechanisms underlying endometrial receptivity. We hypothesise that endometrial metabolism of oestrogen is altered in the supposed embryo-receptive endometria (women with clinical pregnancy) compared to endometria without implantation (women who did not conceive).

Materials and methods

Ethical approval

The Medical Ethics Committee of the University Medical Centre of Utrecht (UMCU) approved the study (registration number 15-495/D), and the study is registered at the Dutch Trial Registry (registration number NL5193/NTR5342; https://trialsearch.who. int/Trial2.aspx?TrialID=NTR6687). Written informed consent was obtained from all participating subjects in accordance with the Helsinki Declaration.

Study design and participants

In this prospective cohort study, endometrial tissue samples were obtained from the SCRaTCH study, a randomised controlled trial investigating the effect of endometrial scratching in women with implantation failure after a first unsuccessful IVF/ICSI cycle on subsequent live birth rate (van Hoogenhuijze et al. 2021). The full study protocol was published previously (van Hoogenhuijze et al. 2017). Briefly, women aged 18-44 years indicated for IVF/ICSI and who had undergone a previous first full IVF/ICSI cycle, in which all fresh and frozen embryos were transferred but which did not result in clinical pregnancy, were included. Women could have primary infertility, i.e. no pregnancy in the past, or secondary infertility, i.e. women who were pregnant (regardless of pregnancy outcome or method of conception) at least once in the past, but are now unable to become pregnant.

Endometrial biopsies and follow-up

Patients allocated to the intervention ('endometrial scratching') underwent an endometrial biopsy during the luteal phase of the menstrual cycle prior to the



second IVF/ICSI cycle. From cycle day 10 onwards, urine ovulation tests based on the detection of the luteinising hormone (LH) surge were performed twice a day at home. The biopsy was obtained 5–8 days after a positive urine ovulation test using a Pipelle catheter (CCD International, France). Tissue was stored at –80°C. After endometrial biopsy, patients continued their second IVF/ICSI cycle. During this second cycle, thus after study randomisation and biopsy, endometrial thickness at the time of planning the ovum pick-up, embryo quality and pregnancy results were recorded.

Sample selection and matching

A total of 141 endometrial biopsies were harvested for the SCRaTCH study in six Dutch centres. For the present study, to avoid potential biases, specific inclusion/exclusion criteria were applied to select subjects: patients had to be below 38 years and had to have a fresh single embryo transfer (ET) in the second IVF/ICSI cycle. Patients who did not have a fresh embryo transfer or who miscarried, were excluded. Women were further matched for age, embryo quality and fertility treatment indication. Following the matching procedure, 30 samples were selected to compare intra-tissue steroid levels and divided into two clinical groups based on the results of the second IVF/ICSI cycle: 15 women with a clinical pregnancy (which all resulted in a live birth) versus 15 non-pregnant women with a negative urine b-hCG (beta human chorionic gonadotrophin) 14 days after ET (the 'pregnant group' and 'non-pregnant group', respectively). In our primary analysis, we compared women who achieved a clinical pregnancy to those who did not. In a secondary analysis, we compared types of infertility: primary versus secondary infertility.

Synthesis

of 2-(4-carboxy-phenyl)-5,6-dimethylbenzimidazole

The synthesis of 2-(4-carboxyphenyl)-5,6-dimethylbenzimidazole (CDB) was adapted from the procedure reported by Katayama et al. (Katayama et al. 4,5-dimethyl-O-phenylenediamine 1991). Briefly, (5.1 g, 37.8 mmol, 1.00 equiv.) was dissolved in 160 mL ethanol in a 500 mL round-bottomed flask in an ice bath with a magnetic stir bar. A suspension of 4-formylbenzoic acid (5.8 g, 38.6 mmol, 1.02 equiv.) in 160 mL ethanol was added dropwise over the course of 1 h. After 1 h, the round-bottomed flask was the

fitted with a condenser, transferred to an oil bath, and refluxed for 2 h. After cooling, an orange precipitate was allowed to develop in the fridge over 48 h, filtered, rinsed with 2× 50 mL cold ethanol, and then transferred to a conical flask. Subsequently, the washed precipitate was refluxed in a 1:1 water–methanol solution for 1 h, filtered, and then recrystallised from a minimum of boiling ethanol three times. Finally, the resulting solid was dried under vacuum for 48 h prior to NMR analysis. NMR spectra were acquired using a Bruker Avance III HD 700 MHz spectrometer equipped with a cryogenically cooled three-channel TCI probe at 299.7 K. Analysis of these spectra confirming the structure of CDB can be found in Supplementary file 1 (see section on supplementary materials given at the end of this article).

Activity of HSD17B enzymes

Enzyme activity of reductive and oxidative HSD17Bs was determined in endometrial tissue extracts using high-performance liquid chromatography (HPLC) as previously described (Delvoux *et al.* 2007) with minor modifications. All measurements were performed in duplicate. Endometrial tissue was homogenised using a Potter homogenisation device in 1.5 mL Eppendorf tubes in RIPA buffer (50 mM Tris–HCl, pH 7.4, 1% NP-40, 0.25% sodium deoxycholate, 150 mM NaCl, 1 mM EDTA). Tissue homogenates were cleared by centrifugation. Protein concentration was determined by a BC Assay Protein Quantitation Kit (Uptima-Interchim, Montlucon, France).

The conversion of oestrone to 17β -oestradiol was carried out as follows: a 500 µL reaction mixture containing 3.2 mM NADP+, 9 mM glucose 6-phosphate, 1 U glucose-6-phosphate dehydrogenase, 3.3 mM magnesium chloride and 10 nmol oestrone in 50 mM potassium phosphate buffer (pH 7.4) was incubated with 24 to 232 µg of the protein lysate at 37°C for 16 h. The conversion of 17 β -oestradiol into oestrone was measured in a 500 µL reaction mixture containing 5 mM NADP+, 10 nmol 17 β -oestradiol in 50 mM sodium phosphate buffer (pH 7.4) and 24 to 232 µg of the protein lysate incubated at 37°C for 16 h.

To determine the oestrogen formation in the reaction mixture we used a HPLC method. After adding the internal standard (IS) butyl-4-hydroxybenzoate (300 ng), 2 mL of H₂O and 2.5 mL of chloroform to the reaction mixture, the oestrogens were extracted in the organic phase. The chloroform phase was evaporated under nitrogen at 45° C and oestrogens were derivatised with 250 µL CDB work and 250 µL



1-(3-dimethylaminopropyl)-3-ethylcarbodiimide hydrochloride (EDC) solution at 50°C for 20 min. The 500 μ L reaction mix were diluted five times in acetonitrile and 10 μ L were injected in the HPLC system.

The HPLC system (Shimadzu LC-10AD setup) consisted of a SIL-10ADvp autosampler, FCV-10ALvp gradient mixer, LC10Avp pump with high pressure gradient mixing extension, DGU-14A degasser and an RF-10Axl fluorescence detector (excitation wavelength 336 nm, emission wavelength 440 nm). Labsolutions software (Shimadzu) was used for instrument control, data acquisition and calculation of peak areas. The derivatised steroids were injected into a LiChroCART 250-4 RP 18 column (Merck) and eluted with a gradient of methanol-H₂O (80:20; v/v) at time 0 to methanol-water (90:10; v/v) at time 20 min. The flow rate was set to 0.8 mL/min.

HSD17B1 inhibition

Reactions were carried out with or without the presence of the specific HSD17B1 inhibitor 3-(15b-estronyl)-*N*-(5-methyl-thiazol-2-yl)-propionamide at 5 μ M concentration, also described as compound 21 (Messinger *et al.* 2009). This HSD17B1 inhibitor is an oestrone substitute at the 15th carbon position and has previously been characterised as a potent and specific inhibitor of HSD17B1 with little affinity towards other HSD17Bs and oestrogen receptor alpha (ERa) (Saloniemi *et al.* 2010, Delvoux *et al.* 2014). The compound was gifted by Dr. Koskimies (Forendo Pharma Finland) and was previously used and tested by our team both *in vitro* and *in vivo* (Delvoux *et al.* 2014, Konings *et al.* 2018*b*, Xanthoulea *et al.* 2021, Poirier *et al.* 2022).

The inhibition of the HSD17B1 enzyme was computed as percentage of inhibition (or inhibition of oestrone to 17β -oestradiol conversion) by the formula: 100% – (formation of 17β -oestradiol with inhibitor/ formation of 17β -oestradiol without inhibitor × 100%).

RNA extraction and sequencing

The RNA data are part of a larger data set (the SCRaTCH-2 study), and RNA extraction, sequencing and analysis of data (with a different sample selection) was previously described (Stevens Brentjens *et al.* 2023). The data described in the current article uses the same data set but has a different sample selection. Briefly, RNA was extracted using QIAsymphony SP with the QIAsymphony

RNA kit (according to the Qiagen protocol) and the Illumina TruSeq Stranded mRNA kit was used according to the Illumina protocol. Trimming, alignment, counting and count matrix processing has been previously described (Stevens Brentjens *et al.* 2023). To identify the differentially expressed genes (DEGs), gene count matrix was analysed using DESeq2 (v1.34.0). The false discovery rate (FDR)-adjusted P < 0.05 and log fold change >0.5 were selected as the screening threshold for DEG comparison. The level of expression of 46 genes controlling the local steroid metabolism was analysed (Supplementary Table 1).

For three patients, data on LH timing were missing. Predictive mean matching (PMM) from M.I.C.E. (v. 3.16.0) was used for the imputation of the variable corresponding to LH timing using existing values from the complete cases matched with respect to Bayesian B and stochastic matching distance (van Buren & Groothuis-Oudshoorn 2011). Four samples were excluded due to low RNA integrity number or RNA concentration. Gene ontology analysis was performed on all genes ($P \le 0.05$).

Statistical analysis

Enzyme activity data were analysed using the Wilcoxon–Mann–Whitney rank sum test. Statistical significance was defined as a P < 0.05. For transcriptome, statistical analyses were performed using R (version 4.2.2, R Foundation for Statistical Computing, Vienna, Austria). Statistical analyses of oestrogen metabolism related genes expression (see Supplementary Table 1) were performed using an unpaired two-sample Wilcoxon test adjusted using Bonferroni's correction. Correlations were considered significant with an FDR-adjusted P < 0.05, asymptotic two-tailed *P*-values estimated from Pearson's coefficients.

Results

Main characteristics of the patients and IVF/ ICSI treatments

We explored whether endometrial oxidative and reductive HSD17B activity during the WOI differed in women who achieved a clinical pregnancy (n=15) ('pregnant group') compared to women who did not (n=15) ('non-pregnant group') after fresh single



cleavage-stage embryo transfer. These women were enrolled in the SCRaTCH study and represent a selection of the larger study population (n=141), as described in the 'Materials and Methods' section. There were no relevant statistically significant differences in patient characteristics and treatment characteristics between the study groups (see Supplementary Table 2A and B for the comprehensive data of all participants). Endometrial thickness on the day of the endometrial biopsy was unknown. The endometrial thickness in the subsequent cycle, on the day of ovulation triggering before the fresh embryo transfer, was significantly higher in the pregnant group (Table 1).

Oxidative and reductive HSD17B activity

Activity of the HSD17Bs responsible for the reduction of oestrone to 17β -oestradiol was detected in all samples via HPLC following the derivatisation of oestrogens with CDB (synthesis and characterisation are described in the 'Materials and Methods' section). Formation of 17β -oestradiol ranged from 55 to 1864 pmol/mg protein/h. Median 17β -oestradiol formation in pregnant women was 1053.68 pmol/mg protein/h (interquartile range (IQR): 591.84–1378.47) which was not significantly different compared to non-pregnant women (996.93 pmol/mg protein/h, IQR: 948.03–1302.64), P=0.97 (Fig. 1A).

The activity of the enzymes responsible for the oxidation of 17 β -oestradiol into oestrone ranged from 32 to 1731 oestrone formed/mg protein/h. Median oestrone formation in pregnant women was not significantly different compared to non-pregnant women (737.06 pmol/mg protein/h, IQR: 334.05–1034.22 vs 623.63 pmol/mg protein/h, IQR: 462.26–876.04, respectively, *P*=0.90) (Fig. 1B). The ratio of reductive versus oxidative HSD17B had a median of 1.63 (IQR: 1.12–2.00) in the pregnant woman compared to 1.95 (IQR: 1.38–2.44) in the group of non-pregnant woman, which was not significantly different (*P*=0.57) (Fig. 1C).

Inhibition of HSD17B1

Since HSD17B1 is the most relevant enzyme in the conversion of oestrone to 17β -oestradiol *in vivo*, and it was specifically shown to be involved human embryo implantation (Li *et al.* 2003), we further explored the engagement of HSD17B1 in the measured reductive

Table 1 Patient and IVF/ICSI cycle characteristics of the participants. Data are presented as mean (±s.p.) or number (%).

	Pregnant (<i>n</i> = 15)	Non-pregnant (<i>n</i> = 15)	Р
Age at inclusion – years	33.5 (3.5)	33.8 (3.1)	0.821 ^f
BMI – kg/m ² c	24.43 (4.45)	23.15 (2.41)	0.346 ^f
Female smokers	2 (13.3%)	2 (13.3%)	1.000 ^g
Primary infertility ^a	7 (46.7%)	5 (33.3%)	0.445g
Duration of infertility (months)	38.7 (17.4)	33.6 (16.7)	0.422 ^f
Indication for IVF or ICSI			0.394g
Male	10 (66.7%)	7 (46.7%)	
Idiopathic	5 (33.3%)	7 (46.7%)	
Unilateral tubal pathology	0 (0%)	1 (6.7%)	
Number of embryos transferred before biopsyb			
Fresh	1	1	
Frozen	2.5 (0.9)	2.91 (1.4)	0.421 ^f
Days between LH surge and biopsy ^d	6.5 (0.8)	6.9 (1.0)	0.189 ^f
Endometrial thickness at last ultrasound before ovulation triggering – mm ^c	12.0 (3.1)	9.5 (2.1)	0.018 ^f
Embryo quality of fresh ET ^e after biopsy			
Days after ovum pick-up	2.7 (0.5)	2.9 (0.4)	0.379 ^f
Number of cells	7.27 (2.3)	7.13 (1.5)	0.849 ^f
Fragmentation			0.183g
0-10%	13 (86.7%)	9 (60%)	
10–20%	1 (6.7%)	5 (33.3%)	
20–50%	1 (6.7%)	1 (6.7%)	

^aPrimary infertility: female has never conceived before; ^bThe number of embryos before randomisation for the SCRaTCH trial; ^cData were missing for one participant in the pregnant group; ^dData were missing for two participants in the pregnant group and one participant in non-pregnant group; ^eET, embryo transfer; ^fBy Independent samples *t*-test; ^gBy chi-square test.

© 2023 the author(s) Published by Bioscientifica Ltd



activity by using a specific HSD17B1 inhibitor. A trend was shown when comparing inhibition of HSD17B1 in the pregnant and non-pregnant group albeit non-significant (17.8% vs 27.5%, P=0.059) (Fig. 2A).

As the endometrial contribution to implantation failure might be different in primary infertile women (who never had a pregnancy before) compared to secondary infertile women (where the endometrium showed to be recipient in previous pregnancies), and previous studies with these samples found more pronounced differences between these groups (Bui et al. 2023, Stevens Brentjens et al. 2023), we performed a subgroup analysis with respect to fertility status (primary vs secondary infertility). When only primary infertile women (n=12) were considered, inhibition was significantly decreased in the pregnant group (n=7) compared to the non-pregnant group (n=5)(16.2% vs 27.1%, P=0.042) (Fig. 2B). The activity of HSD17B1 was not significantly different between the pregnant primary infertile and non-pregnant primary infertile group (1029 pmol 17β-oestradiol/mg protein/h vs 953 pmol 17 β -oestradiol/mg protein/h, P = 0.23).

Transcriptome analysis in pregnant and nonpregnant groups

RNA-sequencing data were available for 26 of the 30 patients (pregnant, n = 14 and non-pregnant, n=12). Four samples were excluded due to low RNA quality or concentration. We first analysed the expression levels of the 46 genes encoding for the enzymes controlling the local steroid metabolism (Supplementary Table 1) (Konings et al. 2018a), and 11 genes out of the 46 (i.e. AKR1C4, AKR1D1, ARO, CYP11B1, CYP11B2, CYP17A1, HSD11B1, HSD3B1, HSD3B2, SULT1B1 and SULT2A1) were not detected. In the 35 detectable transcripts, genes encoding for steroid hormone receptors (ESR1 and PGR), for the androgenmetabolising enzyme SRD5A3 and for hydroxysteroiddehydrogenases HSD11B2 and HSD17B4 showed the highest expression levels. No genes were differentially expressed between the study groups, although a trend was observed for HSD17B13 (adjusted P=0.075) when comparing pregnant (n=14) and non-pregnant (n=12)women (Fig. 3A).



Figure 1 Activity of (A) reductive and (B) oxidative HSD17Bs and (C) activity ratio of the reductive and oxidative HSD17Bs. The (A) conversion of oestrone to 17β -oestradiol, (B) conversion of 17β -oestradiol to oestrone and (C) the activity ratio of reductive and oxidative HSD17Bs was similar in the pregnant (n = 15) and non-pregnant (n = 15) group. Box plots show the 25th and 75th percentile within the median, and whiskers indicate 1.5 times the interquartile range. Statistical significance was determined using a Wilcoxon–Mann–Whitney rank sum test.

© 2023 the author(s) Published by Bioscientifica Ltd



This work is licensed under a Creative Commons destributional contraction of the contract of t



Figure 2 Inhibition of HSD17B1 in the complete study group (A) and subgroup of primary infertile women (B). The inhibition of HSD17B1 was similar in the pregnant (n = 15) and non-pregnant (n = 15) group (A). However, when only women with primary infertility were considered, inhibition of HSD17B1 was significantly lower in the pregnant group (n = 7) compared to the non-pregnant group (n = 5, P = 0.042) (B). Box plots show the 25th and 75th percentile within the median, and whiskers indicate 1.5 times the interquartile range. Statistical significance was determined using a Wilcoxon–Mann–Whitney rank sum test; *P < 0.05.

Next, we analysed the complete transcriptome and four DEGs were identified: HBA1, HBB, GDF15 and *CDKL1* (adjusted P < 0.05) (Table 2 and 3). Characteristics of the expression profiles inspected through a principal component analysis (PCA) did not separate the pregnant (n=14) and non-pregnant (n=12) groups (Supplementary Fig. 1A). Likewise, the PCA plot from the expression matrix of steroid-metabolising genes only did not distinguish between the two groups (Supplementary Fig. 2). Neither the activity nor the inhibition of HSD17B1 was correlated with HSD17B1 expression (Supplementary Fig. 3A and B). Correspondingly, activity of HSD17B2 was not correlated with HSD17B2 expression (Supplementary Fig. 3C). Although the sample size is small, the expression of HSD17B1 was correlated with the expression of HSD17B5, and the expression of HSD17B7 was correlated with the expression of HSD17B1, 2 and 5 in the non-pregnant group, but no correlation

Reproduction

L B P M Stevens Brentjens

© 2023 the author(s) Published by Bioscientifica Ltd could be seen in the pregnant group (Supplementary Fig. 3A and B).

Transcriptome analysis in relation to infertility type

As the differences in 17BHSD1 inhibition were more pronounced in primary infertile women (n=12)we performed a subgroup analysis with respect to infertility type (primary infertility, n=10 vs secondary infertility, n=16, irrespective of the outcome of the second cycle). Data from four RNA samples were missing due to low RNA quality or concentration (primary infertility, n=2, secondary infertility, n=2). When analysing genes encoding for steroid-metabolising enzymes irrespective of pregnancy status, no significantly different expression was found (Fig. 4). PCA did not separate the primary and secondary infertile groups (Supplementary Fig. 1B).





Figure 3 Expression of steroid metabolism related genes by RNA sequencing. Expression of 34 steroid metabolism related genes in the pregnant (n = 12) and non-pregnant (n = 14) group. Box plots show the 25th and 75th percentile within the median, and whiskers indicate 1.5 times the interguartile range. Statistical significance was determined using an unpaired two-sample Wilcoxon test adjusted using Bonferroni's correction; *Putative steroidmetabolising action.

When considering only primary infertile women (n=10) transcriptome analysis revealed 12 DEGs between pregnant (n=6) and non-pregnant (n=4)primary infertile women, although PCA analysis did not show separation of the groups (Supplementary Fig. 1C). The 12 DEGs were FLVCR1-DT, RASAL2-AS1, FAM225B,

IGKC, JCHAIN, IGHGP, USP6, ALOX15B, IGHM, IGHG1, LINC02349, and IGHG3 (Table 2 and 3). Gene ontology analysis indicated two enriched gene ontology biological processes categories: humoral immune response and defence response to bacterium. No steroid-metabolising enzyme was present in the DEGs (Fig. 5).

This work is licensed under a Creative Commons own' \$ ĊCC

Table 2	Differentially expressed gene (DEG) analysis betweer
the pregr	hant ($n = 14$) and non-pregnant ($n = 12$) group ^a .

Gene name	Log2fold change	Log fold change SE	P	Adjusted P
HBA1	1.975	0.433	5.23 × 10 ⁻⁰⁶	0.026
HBB	1.960	0.425	4.03 × 10 ⁻⁰⁶	0.026
GDF15	1.127	0.241	3.02 × 10 ⁻⁰⁶	0.026
CDKL1	-0.567	0.115	8.81 × 10 ⁻⁰⁷	0.017

 $\ensuremath{^{\mathrm{a}}\text{Data}}$ were missing from four participants due to low RNA quality or concentration.

Discussion

In this study, we compared the activity of oxidative and reductive HSD17B enzymes in combination with the endometrial transcriptome between pregnant and non-pregnant women after a fresh ET. The activity of the oxidative and reductive HSD17Bs did not differ between the two patient groups, but in primary infertile patients, inhibition of HSD17B1 was significantly decreased in pregnant compared to non-pregnant women. In addition, among primary infertile patients, 12 DEGs were found between pregnant and non-pregnant women.

To our knowledge, this is the first study that explores oestrogen-metabolising enzyme activity in the endometrium of pregnant and non-pregnant IVF/ ICSI patients. Previous studies have explored enzyme activity and HSD17B1 inhibition in both healthy endometrial tissue and women with endometriosis (Delvoux *et al.* 2009, 2014). Here, the activity of HSD17Bs converting oestrone into 17β -oestradiol was

Table 3 Differentially expressed gene (DEG) analysis between the pregnant (n = 6) and non-pregnant (n = 4) group for primary infertile women only^a.

Gene name	Log2fold change	Log fold change SE	P	Adjusted P
FLVCR1-DT	1.912	0.450	2.17 × 10 ⁻⁰⁵	0.047
RASAL2-AS1	-1.439	0.330	1.30 × 10 ⁻⁰⁵	0.034
FAM225B	-1.653	0.342	1.34 × 10 ⁻⁰⁶	0.007
IGKC	-2.690	0.556	1.29 × 10 ⁻⁰⁶	0.007
JCHAIN	-2.810	0.615	4.85 × 10 ⁻⁰⁶	0.021
IGHGP	-2.903	0.673	1,60 × 10 ⁻⁰⁵	0.038
USP6	-3.026	0.693	1,25 × 10 ⁻⁰⁵	0.034
ALOX15B	-3.177	0.719	1,00 × 10 ⁻⁰⁵	0.033
IGHM	-3.419	0.708	1,37 × 10 ^{–06}	0.007
IGHG1	-3.790	0.6675	1.34 × 10 ⁻⁰⁸	<0.000
LINC02349	-4.100	0.922	8.81 × 10 ⁻⁰⁶	0.033
IGHG3	-4.919	0.965	3.42 × 10 ⁻⁰⁷	0.004

 $\ensuremath{^{\mathrm{s}}}\xspace{\mathrm{Data}}$ were missing from four participants due to low RNA quality or concentration.

https://raf.bioscientifica.com	
https://doi.org/10.1530/RAF-23-0065	

© 2023 the author(s) Published by Bioscientifica Ltd found to be higher in endometriosis tissue compared to the endometrium inside the uterus. Similar to our results, formation of 17β -oestradiol could not always be completely blocked using an HSD17B1-specific inhibitor, indicating that other reducing HSD17Bs may be involved in the generation of 17β -oestradiol in the endometrium (Konings *et al.* 2018*a*).

Four genes were differentially expressed between pregnant and non-pregnant women. *GDF15*, which encodes for a secreted ligand of the transforming growth factor beta superfamily of proteins, is involved in decidualisation, placentation and pregnancy. In a study with pregnant rats, *GDF15* expression was increased at the implantation site, suggesting a role in embryo implantation (Zhao *et al.* 2016). Furthermore, in addition to secretion of hCG, the pregnancy-specific glycoprotein GDF15 was detected in the secretome of human trophoblast organoids (Turco *et al.* 2018). In our analysis, the expression of *GDF15* was higher in the pregnant group.

More pronounced differences in the transcriptome were found in primary infertile women. Among the DEGs specifically identified in this subgroup analysis, LINC02349 has been linked to endometrial receptivity associated pathways before. This long non-coding RNA (IncRNA) may play crucial roles in a series of endometrial events before or during implantation by targeting the Wnt pathways (Tepekoy et al. 2015). In a study using transcriptome data of endometrial tissues from recurrent implantation failure (RIF) patients and fertile controls (data accession number GSE26787) to determine biomarkers for RIF, LINC02349 was identified as one of the eight differentially expressed IncRNAs that distinguish RIF from fertile women. In line with our results, the expression of LINC02349 was downregulated in RIF women compared to fertile controls (Lin & Lin 2022).

One study explored the endometrial intracrinology during IVF/ICSI treatment. Oocyte donors were treated with a standard IVF/ICSI antagonist protocol including ovarian hyperstimulation with folliclestimulating hormone, and an endometrial biopsy was conducted in the putative WOI. Among other genes, the expression of *HSD17B2* was lower compared to control subjects in a natural menstrual cycle, indicating that in stimulated cycles, where the endometrium is exposed to supraphysiological steroid hormone levels, alterations in steroid-metabolising enzymes are induced (Vani *et al.* 2007). In our study, biopsies were taken in the cycle preceding the IVF/ICSI cycle, and no differences





Figure 4 Expression of steroid metabolism related genes by RNA sequencing. Expression of 34 steroid metabolism related genes in primary infertile (n = 10) and secondary infertile (n = 16) women. Box plots show the 25th and 75th percentile within the median, and whiskers indicate 1.5 times the interquartile range. Statistical significance was determined using an unpaired two-sample Wilcoxon test adjusted using Bonferroni's correction; *Putative steroid-metabolising action.

in *HSD17B2* expression were found between pregnant and non-pregnant IVF/ICSI patients.

One strength of our study is its prospective design, where biopsies were obtained as part of a randomised

controlled trial in well-defined patient groups. The biopsy was precisely timed between 5 and 8 days after the LH surge and the transcriptomic data was corrected for differences in LH timing accordingly.

Reproduction © Fertility





Figure 5 Expression of steroid metabolism related genes by RNA sequencing. Expression of 34 steroid metabolism related genes in the pregnant (n = 4) and non-pregnant (n = 6) group considering women with primary infertility only. Box plots show the 25th and 75th percentile within the median, and whiskers indicate 1.5 times the interquartile range. Statistical significance was determined using an unpaired two-sample Wilcoxon test adjusted using Bonferroni's correction; *Putative steroid-metabolising action.

It should be noted, however, that ovulation was not confirmed by serial ovarian ultrasound monitoring. Two limitations arise from the invasive nature of endometrial tissue sampling and the corresponding detrimental effect of sampling in the cycle of the ET: (i) the tissue is scarce, which has led to a smaller study sample size. This could be improved by larger future studies with better differentiation between the causes of infertility, and (ii) potential intercyclic variation bias, since the biopsy has to be taken in the cycle preceding



the embryo transfer cycle. Consequently, a pregnancy in the cycle after the biopsy might not always reflect a receptive endometrium during the cycle of the biopsy. Finally, embryo quality has been taken into account based on morphological embryo criteria. However, aneuploidy status of the embryo was unknown as pre-implantation genetic testing for aneuploidy is currently prohibited in the Netherlands. The discovery of 12 DEGs between pregnant and non-pregnant primary infertile patients is interesting, but a recent publication by Walker et al. has pointed out high inconsistencies in reported DEGs between studies on the endometrial transcriptome (Walker et al. 2023). The current study is limited by the fact that the findings could not be validated with another technique such as real-time PCR, as no surplus samples were available. For a possible translation to relevant biomarkers or druggable pathways, future research with validation in independent larger sample sets or studies with stricter selection criteria for reproductive failure are warranted. For instance, only women with primary infertility or women with RIF could be considered instead of the currently included women with a first failed IVF/ICSI cycle. Besides, non-invasive methods for endometrial assessment in the cycle of the transfer such as a liquid biopsy (i.e. sampling endometrial fluid) could eliminate intercyclic variation in the future (Schobers et al. 2021, Stevens Brentjens et al. 2022).

Conclusions

In conclusion, endometrial intracrinology involves intricate and highly complex pathways that generate steroid metabolites with various activities. participation The of other steroid-metabolising enzymes and the various steroid metabolites including progestogens, androgens and corticosteroids was outside of the scope of this study, but their role in receptivity and implantation in the same study population is described elsewhere (Stevens Brentjens et al. 2023). To the best of our knowledge, this pilot study is the first to explore the endometrial oestrogen balance in humans during implantation by using an unique study population. The study results, especially the role of other HSD17Bs (such as 5, 7 and 12) in pregnancy, need to be reconfirmed in future studies, as this study was limited by a small sample size given its exploratory and pilot nature.

Supplementary materials

This is linked to the online version of the paper at https://doi.org/10.1530/ RAF-23-0065.

Declaration of interest

The authors declare that there is no conflict of interest that could be perceived as prejudicing the impartiality of the study reported.

Funding

The randomised controlled trial (SCRaTCH) was funded by the Netherlands Organisation for Health Research and Development, 'ZonMw' (ZonMw project number 843002601). The current project was funded by the Executive Board of Maastricht University and Maastricht University Medical Centre+. The project is supported by the EVA (Erfelijkheid Voortplanting & Aanleg) specialty program (grant No. KP111513) of Maastricht University Medical Centre (MUMC+), Horizon 2020 innovation grant (ERIN, grant no. EU952516), and the Horizon Europe (NESTOR) (Grant No. 101120075) of the European Commission.

Author contribution statement

Study design: LSB, AR, BD, FB. Tissue collection: LSB, JdH, RvG, Fe.M, JdB, GT. Data collection: DO, BD, MZE, Fr.M, MB, BB. Data analysis: LSB, BD, DO, MZE, AR. Manuscript writing: LSB, AR. Manuscript review and approval: all authors.

Acknowledgements

The authors are grateful to the participants who donated tissue for research.

References

- Bui BN, Van Hoogenhuijze N, Viveen M, Mol F, Teklenburg G, De Bruin JP, Besselink D, Brentjens LS, Mackens S, Rogers MRC, *et al.* 2023 The endometrial microbiota of women with or without a live birth within 12 months after a first failed IVF/ICSI cycle. *Scientific Reports* **13** 3444. (https://doi.org/10.1038/s41598-023-30591-2)
- Das A, Mantena SR, Kannan A, Evans DB, Bagchi MK & Bagchi IC 2009 De novo synthesis of estrogen in pregnant uterus is critical for stromal decidualization and angiogenesis. *PNAS* **106** 12542–12547. (https://doi.org/10.1073/pnas.0901647106)
- Delvoux B, Husen B, Aldenhoff Y, Koole L, Dunselman G, Thole H & Groothuis P 2007 A sensitive HPLC method for the assessment of metabolic conversion of estrogens. *Journal of Steroid Biochemistry and Molecular Biology* **104** 246–251. (https://doi.org/10.1016/j. jsbmb.2007.03.006)
- Delvoux B, Groothuis P, D'Hooghe T, Kyama C, Dunselman G & Romano A 2009 Increased production of 17beta-estradiol in endometriosis lesions is the result of impaired metabolism. *Journal* of Clinical Endocrinology and Metabolism **94** 876–883. (https://doi. org/10.1210/jc.2008-2218)
- Delvoux B, D'hooghe T, Kyama C, Koskimies P, Hermans RJ, Dunselman GA & Romano A 2014 Inhibition of type 1 17beta-hydroxysteroid dehydrogenase impairs the synthesis of 17beta-

estradiol in endometriosis lesions. *Journal of Clinical Endocrinology and Metabolism* **99** 276–284. (https://doi.org/10.1210/jc.2013-2503)

- **DeMayo FJ & Lydon JP** 2020 90 years of progesterone: new insights into progesterone receptor signaling in the endometrium required for embryo implantation. *Journal of Molecular Endocrinology* **65** T1–T14. (https://doi.org/10.1530/JME-19-0212)
- European IVF Monitoring Consortium (EIM) for the European Society of Human Reproduction and Embryology (ESHRE), Wyns C, De Geyter C, Calhaz-Jorge C, Kupka MS, Motrenko T, Smeenk J, Bergh C, Tandler-Schneider A, Rugescu IA, et al. 2022 ART in Europe, 2018: results generated from European registries by ESHRE. Human Reproduction Open. (https://doi.org/10.1093/hropen/ hoaa038)
- Gibson DA, Mcinnes KJ, Critchley HO & Saunders PT 2013 Endometrial Intracrinology-generation of an estrogen-dominated microenvironment in the secretory phase of women. *Journal of Clinical Endocrinology and Metabolism* **98** E1802–E1806. (https://doi. org/10.1210/jc.2013-2140)
- Katayama M, Masuda Y & Taniguchi H 1991 Determination of alcohols by high-performance liquid chromatography after pre-column derivatization with 2-(4-carboxyphenyl)-5,6dimethylbenzimidazole. *Journal of Chromatography* 585 219–224. (https://doi.org/10.1016/0021-9673(91)85080-Y)
- Konings G, Brentjens L, Delvoux B, Linnanen T, Cornel K, Koskimies P, Bongers M, Kruitwagen R, Xanthoulea S & Romano A 2018a Intracrine regulation of estrogen and other sex steroid levels in endometrium and non-gynecological tissues; pathology, physiology, and drug discovery. *Frontiers in Pharmacology* 9 940. (https://doi.org/10.3389/fphar.2018.00940)
- Konings GF, Cornel KM, Xanthoulea S, Delvoux B, Skowron MA, Kooreman L, Koskimies P, Krakstad C, Salvesen HB, Van Kuijk K, et al. 2018b Blocking 17beta-hydroxysteroid dehydrogenase type 1 in endometrial cancer: a potential novel endocrine therapeutic approach. *Journal of Pathology* 244 203–214. (https://doi.org/10.1002/ path.5004)
- Labrie F 1991 Intracrinology. *Molecular and Cellular Endocrinology* 78 C113-C118. (https://doi.org/10.1016/0303-7207(91)90116-a)
- Li Y, Qin L, Xiao ZJ, Wang YL, Herva R, Leng JH, Lang JH, Isomaa V & Piao YS 2003 Expression of P450 aromatase and 17betahydroxysteroid dehydrogenase type 1 at fetal-maternal interface during tubal pregnancy. *Journal of Steroid Biochemistry and Molecular Biology* 87 241–246. (https://doi.org/10.1016/j.jsbmb.2003.09.013)
- Lin N & Lin JZ 2022 Identification of long non-coding RNA biomarkers and signature scoring, with competing endogenous RNA networkstargeted drug candidates for recurrent implantation failure. *Human Fertility* **25** 983–992. (https://doi.org/10.1080/14647273.2021.1956693)
- Messinger J, Husen B, Koskimies P, Hirvela L, Kallio L, Saarenketo P & Thole H 2009 Estrone C15 derivatives-a new class of 17betahydroxysteroid dehydrogenase type 1 inhibitors. *Molecular and Cellular Endocrinology* **301** 216–224. (https://doi.org/10.1016/j.mce.2008.10.022)
- Poirier D, Nyachieo A, Romano A, Roy J, Maltais R, Chai D, Delvoux B, Tomassetti C & Vanhie A 2022 An irreversible inhibitor of 17beta-hydroxysteroid dehydrogenase type 1 inhibits estradiol synthesis in human endometriosis lesions and induces regression of the non-human primate endometriosis. *Journal of Steroid Biochemistry and Molecular Biology* 222 106136. (https://doi. org/10.1016/j.jsbmb.2022.106136)
- Saloniemi T, Jarvensivu P, Koskimies P, Jokela H, Lamminen T, Ghaem-Maghami S, Dina R, Damdimopoulou P, Makela S, Perheentupa A, et al. 2010 Novel hydroxysteroid (17beta) dehydrogenase 1 inhibitors reverse estrogen-induced endometrial hyperplasia in transgenic mice. American Journal of Pathology 176 1443–1451. (https://doi.org/10.2353/ajpath.2010.090325)
- Schobers G, Koeck R, Pellaers D, Stevens SJC, Macville MVE, Paulussen ADC, Coonen E, van den Wijngaard A, de Die-Smulders C, de Wert G, *et al.* 2021 Liquid biopsy: state

of reproductive medicine and beyond. *Human Reproduction* **36** 2824–2839. (https://doi.org/10.1093/humrep/deab206)

- Stevens Brentjens LBPM, Habets D, Den Hartog J, Al-Nasiry S, Wieten L, Morre S, Van Montfoort A, Romano A & Van Golde R 2022 Endometrial factors in the implantation failure spectrum: protocol of a MUltidisciplinary observational cohort study in women with Repeated Implantation failure and recurrent Miscarriage (MURIM study). *BMJ Open* **12** e056714. (https://doi.org/10.1136/bmjopen-2021-056714)
- Stevens Brentjens LBPM, Obukhova D, den Hartog J, Delvoux B, Koskivuori J, Auriola S, Häkkinen M, Bui B, van Hoogenhuijze N, Mackens S, et al. 2023 An integrative analysis of endometrial steroid metabolism and transcriptome in relation to endometrial receptivity in IVF patients. F&S Science 4 219–228. (https://doi.org/10.1016/j.xfss.2023.04.003)
- Tepekoy F, Akkoyunlu G & Demir R 2015 The role of Wnt signaling members in the uterus and embryo during pre-implantation and implantation. *Journal of Assisted Reproduction and Genetics* **32** 337–346. (https://doi.org/10.1007/s10815-014-0409-7)
- Turco MY, Gardner L, Kay RG, Hamilton RS, Prater M, Hollinshead MS, Mcwhinnie A, Esposito L, Fernando R, Skelton H, et al. 2018 Trophoblast organoids as a model for maternal-fetal interactions during human placentation. Nature 564 263–267. (https://doi.org/10.1038/s41586-018-0753-3)
- van Buren S & Groothuis-Oudshoorn K 2011 Mice: multivariate imputation by chained equations in R. *Journal of Statistical Software* 45. (https://doi.org/10.18637/jss.v045.i03)
- van Hoogenhuijze NE, Torrance HL, Mol F, Laven JSE, Scheenjes E, Traas MAF, van Hoogenhuijze NE, Janssen C, Cohlen B, Teklenburg G, et al. 2017 Endometrial scratching in women with implantation failure after a first IVF/ICSI cycle; does it lead to a higher live birth rate? The SCRaTCH study: a randomized controlled trial (NTR 5342). BMC Women's Health 17 47. (https://doi. org/10.1186/s12905-017-0378-y)
- van Hoogenhuijze NE, Mol F, Laven JSE, Groenewoud ER, Traas MAF, Janssen CAH, Teklenburg G, de Bruin JP, van Oppenraaij RHF, Maas JWM, *et al.* 2021 Endometrial scratching in women with one failed IVF/ICSI cycle-outcomes of a randomised controlled trial (SCRaTCH). *Human Reproduction* **36** 87–98. (https:// doi.org/10.1093/humrep/deaa268)
- Vani S, McDonald SE, Williams AR, Mason JI, Thong KJ & Critchley HO 2007 Mid-luteal endometrial intracrinology following controlled ovarian hyperstimulation involving use of a gonadotrophin releasing hormone antagonist. *Human Reproduction* 22 2981–2991. (https://doi.org/10.1093/humrep/dem269)
- Walker ER, Mcgrane M, Aplin JD, Brison DR & Ruane PT 2023 A systematic review of transcriptomic studies of the human endometrium reveals inconsistently reported differentially expressed genes. *Reproduction and Fertility* 4. (https://doi.org/10.1530/RAF-22-0115)
- Xanthoulea S, Konings GFJ, Saarinen N, Delvoux B, Kooreman LFS, Koskimies P, Hakkinen MR, Auriola S, D'avanzo E, Walid Y, et al. 2021 Pharmacological inhibition of 17beta-hydroxysteroid dehydrogenase impairs human endometrial cancer growth in an orthotopic xenograft mouse model. *Cancer Letters* 508 18–29. (https://doi.org/10.1016/j.canlet.2021.03.019)
- Zhao L, Isayama K, Chen H, Yamauchi N, Shigeyoshi Y, Hashimoto S & Hattori MA 2016 The nuclear receptor REV-ERBalpha represses the transcription of growth/differentiation factor 10 and 15 genes in rat endometrium stromal cells. *Physiological Reports* 4. (https://doi.org/10.14814/phy2.12663)

Received 19 September 2023 Accepted 14 November 2023 Available online 14 November 2023 Version of Record published 13 December 2023



e230065

4:4